Author's response to reviews

Title: Disease burden and related medical costs of rotavirus infections in Taiwan

Authors:

Chun-Yi Lu (chunyi@ha.mc.ntu.edu.tw)
Tsai-Ling Lauderdale (lauderdale@nhri.org.tw)
Yin-Hua Fang (yinna@ha.mc.ntu.edu.tw)
Chung-Yi Wang (tucle91@hotmail.com)
Yu-Huai Ho (hoyuhuai@yahoo.com.tw)
Che-Lun Hung (ly7077@tpts6.seed.net.tw)
Luan-Yin Chang (allen0104.tw@yahoo.com.tw)
Chin-Yun Lee (mfsun@ha.mc.ntu.edu.tw)
Li-Min Huang (lmhuang@ha.mc.ntu.edu.tw)

Version: 10 Date: 9 November 2006

Author's response to reviews:

Dear Editors of the BMC Infectious Diseases:

Enclosed please find the revised manuscript entitled "Disease burden and related medical costs of rotavirus infections in Taiwan" (MS: 1882125533102107). The manuscript has been copyedited for English mistakes as suggested by Reviewer #2. The study has been approved by the Clinical Research Ethics Committee of the National Taiwan University Hospital. A statement has been added to the end of Methods section. The manuscript has also been double checked to make sure it conforms to the journal style.

We look forward to hearing your final decision on the revision. Thank you.

Sincerely yours,

Tsai-Ling Lauderdale
Division of Clinical Research
National Health Research Institutes
35 Keyan Road, Zhunan Town
Miaoli County, Taiwan 350